Title: Follow-up care after treatment for prostate cancer: protocol for an evaluation of a nurse-led supported self-management and remote surveillance programme.

Authors:

Jane Frankland (j.l.frankland@soton.ac.uk)
Hazel Brodie (h.brodie@soton.ac.uk)
Deborah Cooke (d.cooke@soton.ac.uk)
Claire Foster (c.l.foster@soton.ac.uk)
Rebecca Foster (r.m.foster@soton.ac.uk)
Heather Gage (h.gage@surrey.ac.uk)
Jake Jordan (jake.jordan@soton.ac.uk)
Ines Mesa-Eguiagaray (I.Mesa-Eguiagaray@soton.ac.uk)
Ruth Pickering (r.m.pickering@soton.ac.uk)
Alison Richardson (alison.richardson@soton.ac.uk)

Version: 1 Date: 08 Feb 2017

Author’s response to reviews:

Dafne Solera
Executive Editor
BMC Cancer
BiomedCentral

8th February 2017

Ref: BCAN-D-16-02735
Dear Dr Solera

Follow-up care after treatment for prostate cancer: protocol for an evaluation of a nurse-led supported self-management and remote surveillance programme.

Thank you for your email of 27th January 2017 and for your consideration of our manuscript to BMC Cancer. The requested responses are provided below.

1. Ethical and Funding Approval Documentation

Copies of ethical approval and funding approval have been sent to BMCSeriesEditorial@biomedcentral.com as requested in your email.

2. Funding

The project proposal was peer reviewed by the funding body prior to award. The detailed study protocol then underwent independent peer-review at our request.

3. Study status

The study is ongoing. Data collection was completed on 20.12.16 and we are now analysing the data.

4. Related Articles

No manuscripts containing the results of the study have been published or submitted to a journal. The evaluation protocol has been presented as posters at two conferences as detailed below:


We look forward to hearing from you further.

Yours sincerely

Dr Jane Frankland
Senior Research Fellow
Faculty of Health Sciences
University of Southampton
Building 67, Room 4005
Highfield
Southampton
SO17 1BJ